Author's response to reviews

Title: Structural equation and log-linear modeling: a comparison of methods in the analysis of a study on caregivers' health

Authors:

Bin Zhu (bzh@epid.jgh.mcgill.ca)
Stephen D Walter (walter@mcmaster.ca)
Peter L Rosenbaum (rosenbau@mcmaster.ca)
Dianne J Russell (russelld@mcmaster.ca)
Parminder Raina (praina@mcmaster.ca)

Version: 5 Date: 12 October 2006

Author's response to reviews:

Oct 12, 2006

To BioMed Central Medical research methodology

Re: Structural equation and log-linear modeling: a comparison of methods in the analysis of a study on caregivers' health

We have made the changes you requested for the production process.

References - The reference list should contain all authors' names, if there are 30 or below. The term 'et al.' should not be used. See ref 40.

-- We have included full authors list, removed the term "et al" in reference 40.

Figures - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure cropping guide may be of some help, and can be viewed here: http://www.biomedcentral.com/info/ifora/cropping.

-- Figures have been cropped.

Please let me know if you need any further changes.

Sincerely,

Bin Zhu, MSc,
Centre for Clinical Epidemiology & Community Studies
Lady Davis Institute for Medical Research